Author’s response to reviews

Title: Understanding the impact of five major determinants of health (genetics, biology, behavior, psychology, society/environment) on type 2 diabetes in U.S. Hispanic/Latino families: Mil Familias - A cohort study

Authors:

Jessikah Morales (nglantz@sansum.org;jmoralessansum.org;rod.everhart@syneoshealth.com;nglantz@sansum.org)
Namino Glantz (nglantz@sansum.org)
Arianna Larez (alarez@sansum.org)
Wendy Bevier (wbevier@sansum.org)
Mary Conneely (mconneely@sansum.org)
Ludi Fan (ludi.fan@lilly.com)
Beverly Reed (reed_beverly_l@lilly.com)
Carlos Alatorre (alatorre_carlos@lilly.com)
Rosirene Paczkowski (paczkowski_rosirene@lilly.com)
Tamim Ahmed (tahmed@sactuaries.com)
Andrew Mackenzie (amackenzie@sactuaries.com)
Ian Duncan (iduncan@sactuaries.com)
David Kerr (dkerr@sams.org)

Version: 2 Date: 15 Dec 2019

Author’s response to reviews:

We thank the reviewers and editor for their comprehensive and constructive comments. We have addressed each comment, as detailed below.
Editor Comments:

1. Please move the trial registration information in the abstract to a "Trial registration" section in the abstract, as in our submission guidelines https://bmcendocrdisord.biomedcentral.com/submission-guidelines/preparing-your-manuscript/study-protocol.

In this section, please clarify whether the trial was prospectively or retrospectively registered at clinicaltrials.gov. If it was retrospectively registered please note this in your manuscript alongside information on trial registration.


2. Please ensure that all sections in your manuscript adhere to our submission guidelines https://bmcendocrdisord.biomedcentral.com/submission-guidelines/preparing-your-manuscript/study-protocol

COMPLETED.

3. (a) Please upload English language versions of all questionnaires developed specifically for use in this study as supplementary files.

COMPLETED. The study-specific questionnaires (Introductory Interview and demographics, Activity Monitor Survey, Reflection Questionnaire) have been uploaded as supplementary files.

(b) If the questionnaires are not yours or are already published by you, please supply references and/or links to them.

COMPLETED. These are listed in References as items 13-36, pages 27-29, lines 549-608.
(c) Please remember to add these to the supplementary file list, and

COMPLETED. These are listed in the Supplementary Files section, added to the manuscript after the Abbreviations section, pages 22-23, lines 457-460.

(d) add a statement in the Availability of data and materials section referring to them.

COMPLETED. The following statement was added to the Availability of data and materials section, page 23, lines 475-478:

The SDRI-developed study-specific questionnaires (Introductory Interview and Demographics, Activity Monitor Survey, Reflection Questionnaire) have been uploaded as supplementary files. References and/or links to all other study materials are provided in the References section.

4. In the “Funding” section of your declarations, please clarify the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

COMPLETED. See Funding section, page 24, lines 494-499.

5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

COMPLETED. The revised manuscript has been uploaded. The figure, tables, and supplementary files are uploaded as separate files, as specified.